Author's response to reviews

Title: Validity of a self-reported diagnosis of depression among participants in a cohort study using the Structured Clinical Interview for DSM-IV (SCID-I)

Authors:

Almudena Sanchez-Villegas (asanchez@dcc.ulpgc.es)
Javier Schlatter (schlatter@unav.es)
Felipe Ortuno (fortunos@unav.es)
Francisca Lahortiga (fhortiga@unav.es)
Jorge Pla (jpla@unav.es)
Silvia Benito (sbenito@alumni.unav.es)
Miguel A Martinez-Gonzalez (mamartinez@unav.es)

Version: 4 Date: 29 May 2008

Author's response to reviews:

Las Palmas de Gran Canaria, May 29th 2008.

Dear editor,

Please, find attached the manuscript entitled “Validity of a self-reported diagnosis of depression among participants in a cohort study using the Structured Clinical Interview for DSM-IV (SCID-I)”, to be considered by for the editorial board of the BMC Psychiatry for publication.

We have added the formatting requirements.

Sincerely,

Almudena Sanchez-Villegas